Author's response to reviews

Title: The effectiveness of early lens extraction with intraocular lens implantation for the treatment of primary angle-closure glaucoma (EAGLE): study protocol for a randomized controlled trial

Authors:

Augusto Azuara-Blanco (aazblanco@abdn.ac.uk)
Jennifer M Burr (j.m.burr@abdn.ac.uk)
Claire Cochran (claire.cochran@abdn.ac.uk)
Craig Ramsay (c.r.ramsay@abdn.ac.uk)
Luke Vale (luke.vale@newcastle.ac.uk)
Paul Foster (p.foster@ucl.ac.uk)
David Friedman (david.friedman@jhu.edu)
Zahidul Quayyum (z.quayyum@abdn.ac.uk)
Jimmy Lai (laism@hku.hk)
Winnie Nolan (winnie_nolan@yahoo.com)
Tin Aung (Aung.Tin@snec.com.sg)
Paul Chew (ophchewp@nus.edu.sg)
Gladys McPherson (g.mcpherson@abdn.ac.uk)
Alison McDonald (a.mcdonald@abdn.ac.uk)
John Norrie (John.Norrie@glasgow.ac.uk)

Version: 8 Date: 20 May 2011

Author's response to reviews:

The revisions from the have been made. Specifically:

References: References have been cited in the text using consecutive numbers in square brackets, and placed before the full stop.

Authors' contributions: We have added that all authors read and approved the final manuscript.

Author groups: the name of the group has been added to the list of authors, by adding the name to the "first name" box; and the bracketed abbreviation to the "last name" box in the submission system.

The names of the members of the group have been removed from the title page and placed in the Acknowledgements, as they are to be acknowledged as members of the study group.

Tables: we have deleted the colours

Footnotes: We have incorporated the footnotes into the text of the manuscript.